Author's response to reviews

Title: Association between congenital toxoplasmosis and parent-reported developmental outcomes, concerns, and impairments, in 3 year old children.

Authors:

Katherine Freeman (kfreeman@montefiore.org)
Alison Salt (a.salt@ich.ucl.ac.uk)
Andrea Prusa (arprusa@hotmail.com)
Gunilla Malm (gunilla.malm@hs.se)
Nicole Ferret (ferret.n@chu-nice.fr)
Wilma Buffalono (wilma@cds.unina.it)
Dorthe Schmidt (DRS@ssi.dk)
Hooi Kuan Tan (h.tan@ich.ucl.ac.uk)
Ruth E Gilbert (r.gilbert@ich.ucl.ac.uk)
European Multicentre Study on Congenital Toxoplasmosis (emscot) (r.gilbert@ich.ucl.ac.uk)

Version: 5 Date: 15 June 2005

Author's response to reviews: see over
Dear Sir/Madam,

Re: MS1211410730500431 Association between congenital toxoplasmosis and parent-reported developmental outcomes, concerns, and impairments, in 3 year old children. K Freeman, A Salt, A Prusa, G Malm, N Ferret, W Buffalono, D Schmidt, HK Tan and RE Gilbert. BMC Pediatrics

As requested we have now made the following formatting changes as requested in your email of 26th May 2005.

1. Reference 7 has been added to the manuscript following your acceptance of our second article (MS1066325445005647)

2. All references have been formatted in the BMC Pediatrics journal style using the correct medline journal abbreviations for journal names. All author names have been listed for each reference.

This is the final version of the manuscript and has been checked carefully for any typing errors and small changes by the authors.

Yours sincerely

Ruth Gilbert
Reader in Clinical Epidemiology